Author's response to reviews

Title: Cognitive performance and leukocyte telomere length in two narrow age-range cohorts: a population study

Authors:

Karen A Mather (karen.mather@unsw.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)
Kaarin J Anstey (k.anstey@anu.edu.au)
Peter J Milburn (Peter.Milburn@anu.edu.au)
Simon Easteal (s.easteal@anu.edu.au)
Helen Christensen (h.christensen@anu.edu.au)

Version: 9 Date: 6 September 2010

Author's response to reviews:

1. Corrected the spelling of Supplemental on p.15
2. Added a fullstop on p.4
3. Removed an extra space on p.8
4. Corrected the sentence on lines 3 & 4 on p.17.
5. Line added to bottom of Table 2, p.36